Author's response to reviews

**Title:** Optimising intraperitoneal gentamicin dosing in peritoneal dialysis patients with peritonitis (GIPD) study

**Authors:**

Dwarakanathan Ranganathan  
(Dwarakanathan_Ranganathan@health.qld.gov.au)  
Julie M Varghese  
(j.varghese1@uq.edu.au)  
Robert G Fassett  
(r.fassett@uq.edu.au)  
Jeffrey Lipman  
(j.lipman@uq.edu.au)  
Vincent D’Intini  
(Vincent_D'Intini@health.qld.gov.au)  
Helen Healy  
(Helen_Healy@health.qld.gov.au)  
Jason A Roberts  
(j.roberts2@uq.edu.au)

**Version:** 3  **Date:** 7 December 2009

Author's response to reviews:

Formatting changes made as requested- list of changes as outlined in the e-mail from The BioMed Central Editorial Production Team on 30/11/2009 (copied & pasted below).

Formatting changes to be made:

----------------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs).

Abstract: Please note that there must be no space between the letters and numbers of the trial registration number (e.g. ISRCTN12345678). Please ensure that you have also updated the abstract in the submission system to ensure it remains identical to the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Qualifications/Job titles: Please remove the authors' qualifications (e.g. PhD / MD) and job titles from the manuscript and submissions system.

Email address: please use the following format for email addresses, including using authors' initials:

AB: abcd@institution.ac.uk

EF: efgh@generic.co.uk

IJ: IJKL@corporation.com

Abstract: Please rename the methods section 'Methods/Design' and ensure that the manuscript matches exactly what is in the submission system.

Abstract: Please remove the keywords from the abstract.

List of abbreviations: Please rename this section 'Abbreviations'.

Abbreviations: Please format these in to a paragraph.

Tables: the table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

References: Please insert the references directly after the acknowledgements section.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.